Author’s response to reviews

Title: The psychology of "cure" - unique challenges to consent processes in HIV cure research in South Africa

Authors:

Keymanthri Moodley (km@sun.ac.za)
Ciara Staunton (c.staunton@mdx.ac.uk)
Theresa Rossouw (theresa.rossouw@up.ac.za)
Malcolm de Roubaix (malcolmderoubaix@gmail.com)
Zoe Duby (zoe.duby@gmail.com)
Donald Skinner (dskinner@sun.ac.za)

Version: 2 Date: 12 Nov 2018

Author’s response to reviews:

The Editorial Team

Thank you for considering this article for publication. We have responded to the three queries raised by Reviewer 2 as follows:

1. All quotes now reflect the category of person in the singular.

2. In the Table of respondents all categories are indicated in the plural.

3. On page 4 lines 30-38, the methodology is clarified in terms of sample size.

In total, 83 participants were approached for this project. 42 agreed to interviews (IDIs), 15 declined and 26 agreed to Focus Group Discussions (42 + 15 + 26 = 83). Of the 42 IDIs conducted, results of 14 IDIs were published previously. This paper reflects the results of 28 IDIs and 26 participants in FGDs.

There were no comments from Reviewer 1.

Regards

Keymanthri Moodley